Author's response to reviews

Title: Rotational stability of the AcrySof SA60TT toric intraocular lenses: a cohort study

Authors:

Bruno Zuberbuhler (b.zuberbuehler@zubisoft.ch)
Theo Signer (theo.signer@vistaklinik.ch)
Richard Gale (RGALE92@aol.com)
Eduard Haefliger (eduard.haefliger@vistaklinik.ch)

Version: 4 Date: 5 April 2008

Author's response to reviews: see over
Editorial response

RE: BioMed Central, MS: 192053699164865
"Rotational stability of the AcrySof SA60TT toric intraocular lenses: a cohort study"

"I am delighted to say we consider the referees' criticisms satisfactorily addressed, and anticipate acceptance of your revised manuscript. However before acceptance, we request that you make some formatting changes to the manuscript,..."

Dear Editor,

The revised manuscript conforms to the journal style. The following two changes were made:

1. The title page has been changed according to the guidelines provided by you with an Internet link (http://www.biomedcentral.com/bmcophthalmol/ifora/#title). In particular, the ending "a cohort study" was added to the title, the contribution of each author was stated and the e-mail addresses were added.

2. Acknowledgment: There is no specific acknowledgment. No funding was granted to the project.